Author's response to reviews

Title: A cluster-randomised, controlled trial to assess the impact of a workplace osteoporosis prevention intervention on the dietary and physical activity behaviours of working women: study protocol

Authors:

Ai May Tan (amtan@student.unimelb.edu.au)
Anthony D LaMontagne (alamonta@unimelb.edu.au)
Rani Sarmugam (ranics@hotmail.com)
Peter Howard (howardp@unimelb.edu.au)

Version: 2 Date: 6 November 2012

Author's response to reviews: see over